ES 112.02/04 Homework #3 Fall 2003-2004

Date Formatting and Printing

This time your task is to make the computer take a date given in “number” format, as “DD MM YYYY” (e.g., “15 10 2003”), and make it print the same date in “readable” format (e.g., October 15th, 2003).

Sample input output pairs would be as follows:

Number form Readable form 15 10 2003 October 15th, 2003 1 1 2001 January 1st, 2001 31 12 2004 December 31st, 2004 2 3 2000 March 2nd, 2000 11 5 1999 May 11th, 1999

A sample run would look like:

Please enter a date in the form DD MM YYYY: 22 10 2003
October 22nd, 2003